Author's response to reviews

Title: Study protocol of a Dutch smoking cessation e-health program

Authors:

Nicola Stanczyk (nicola.stanczyk@maastrichtuniversity.nl)
Catherine Bolman (catherine.bolman@ou.nl)
Jean W.M. Muris (jean.muris@maastrichtuniversity.nl)
Hein de Vries (hein.devries@maastrichtuniversity.nl)

Version: 2 Date: 26 October 2011

Author's response to reviews: see over
Dear editor,

Attached you will find the revised manuscript ‘Study protocol of a Dutch smoking cessation e-health program’. In this letter we will give a short point-by point response to the concerns of BMC.

1) We registered the trial at the Netherlands Trial register. In the article we included the trial registration number. (see abstract of the manuscript)
2) Furthermore we included an Authors contribution section before the Acknowledgements and Reference list. We briefly described and summarized the author’s contribution to the article and study.
3) Next to this I corrected the figures of the article to the journal style. Furthermore, I revised the reference list, conforming to the journal style.

We look forward to receiving a quick answer and hope that the changes fit to the wishes of the journal.

With best wishes,

Nicola Stanczyk